Author’s response to reviews

Title: Is Lamarckian Evolution Relevant to Medicine?

Authors:

Adam E Handel (ahandel@doctors.org.uk)
Sreeram V Ramagopalan (sreeramr@well.ox.ac.uk)

Version: 4 Date: 10 May 2010

Author’s response to reviews:

Dear sir or madam,

We addressed all of the following points (see below) apart from the conversion of
the abbreviations into a single paragraph.

Best wishes,

Adam & Sreeram

Dr. Adam Handel & Dr. Sreeram Ramagopalan,
Wellcome Trust Centre for Human Genetics,
University of Oxford

Formatting changes to be made:

----------------------------

Major revisions (we require the author to make these changes)

Abstract: please ensure that the Abstract in the manuscript matches exactly what
is in the submission system. At present they are not the same. Please also
remove the wordcount and reference count from the abstract, these are not
necessary.

Authors’ contributions: Please indicate that both authors read and approved the
final manuscript.

~

Minor revisions (we can make these changes for you, although it will speed up
publication of your manuscript if you do them while making the major changes
above)

Title: please use sentence case in the title, both in the manuscript file and the
submission system. Capitalise only the first word and proper nouns, and do not
include full stops/periods. For example, ?The Title of My Prestigious Publication: A Scientific Study.? should be changed to ?The title of my prestigious publication: a scientific study?. Remember to update the submission system to
make sure the title remains consistent.

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submission system.

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Tracking: Please remove all tracked changes and comments from the manuscript, including strikethrough, underlined, highlighted and coloured text, ensuring that only the final version of all text remains. This may involve accepting or rejecting the tracked changes.

Abbreviations: Please format these in to a paragraph.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.